Author's response to reviews

Title: Prevalence of high-risk HPV types and associated genital diseases in women born in 1988/89 or 1983/84 - results of WOLVES, a population-based epidemiological study in Wolfsburg, Germany

Authors:

Karl U Petry (gyn@klinikum.wolfsburg.de)
Alexander Luyten (gyn@klinikum.wolfsburg.de)
Annika Justus (info@conreso.com)
Angelika Iftner (sekretariat.iftner@med.uni-tuebingen.de)
Sarah Strehlke (gyn@klinikum.wolfsburg.de)
Axel Reinecke-Lüthge (Axel.Reinecke-Luethge@Klinikum.Wolfsburg.de)
Renate Schulze-Rath (rschulze-rath@spmsd.com)
Thomas Iftner (sekretariat.iftner@med.uni-tuebingen.de)

Version: 2 Date: 5 January 2013

Author's response to reviews:

The author details have been added as requested